Author's response to reviews

Title: Technology assessment and resource allocation for predictive genetic testing: A study of the perspectives of Canadian genetic health care providers

Authors:

Alethea Adair (aadair@ualberta.ca)
Robyn Hyde-Lay (rhyde-lay@genomealberta.ca)
Edna Einsiedel (einsiede@ucalgary.ca)
Timothy Caulfield (tcaulflid@law.ualberta.ca)

Version: 4 Date: 12 May 2009

Author's response to reviews: see over
May 12, 2009

Natalie Pafitis, M.Sc.
Senior Assistant Editor
BMC-series journals

Re: Technology assessment and resource allocation for predictive genetic testing: A pilot study of the perspectives of Canadian genetic health care providers

We would like to thank-you for accepting our manuscript for publication in BMC Medical Ethics. While we find the reviewer’s comments extremely useful, and certainly recognize the time and effort that was put into reviewing the manuscript, we have chosen to move forward and submit it as if for publication. The only minor revisions made were to correct the corresponding email address of R. Hyde-Lay and the citation number 28.

Reviewer #3:
Minor points:

1) The authors should adapt the e-mail address von RHL since he is now working for Genome Alberta (according to the response letter).
   
   - Please see above

2) Correct citation no 28.
   - Corrected in manuscript